Author's response to reviews

Title: Primary prevention of childhood obesity through counselling sessions at Swedish child health centres: Design, methods and baseline sample characteristics of the PRIMROSE cluster-randomised trial

Authors:

Nora Doering (nora.doering@ki.se)
Lena M Hansson (lena.hansson@ki.se)
Elina Scheers Andersson (elina.scheers.andersson@ki.se)
Benjamin Bohman (benjamin.bohman@ki.se)
Maria Westin (maria.westin@gmail.com)
Margaretha Magnusson (margaretha.magnusson@akademiska.se)
Christel Larsson (christel.larsson@ped.gu.se)
Elionor Sundblom (elionor.sundblom@sll.se)
Mikaela Willmer (mikaela.willmer@ki.se)
Margareta Blennow (margareta.blennow@sodersjukhuset.se)
Berit L Heitmann (berit.lilienthal.heitmann@regionh.dk)
Lars Forsberg (Lars.Forsberg.3@ki.se)
Sanne Wallin (sanna.wallin@bra.se)
Per Tynelius (per.tynelius@ki.se)
Ata Ghaderi (ata.ghaderi@ki.se)
Finn Rasmussen (finn.rasmussen@ki.se)

Version: 3
Date: 27 March 2014

Author's response to reviews: see over
Dear Mr. Hennessy Vargas

We have included the additional information on page 7 line 4 in the manuscript.

We hope to have informed you sufficiently. If you have additional requests or questions, please do not hesitate to contact us again.

Kind regards,

Nora Doering